FebriDx point-of-care test in patients with suspected COVID-19: a pooled diagnostic accuracy study

Samuel G. Urwin¹, B. Clare Lendrem², Jana Suklan², Kile Green², Sara Graziadio¹, Peter Buckle³, Paul M. Dark^{4,5}, Adam L. Gordon^{6,7}, Daniel S. Lasserson⁸⁻¹⁰, Brian Nicholson¹¹, D. Ashley Price¹, Charles Reynard^{12,13}, Mark H. Wilcox¹⁴, Graham Prestwich¹⁵, Valerie Tate¹⁶, Tristan W. Clark¹⁷⁻¹⁹, Raja V. Reddy²⁰, Richard Body^{13,21}, and A. Joy Allen² on behalf of the CONDOR steering group

¹NIHR Newcastle In Vitro Diagnostics Co-operative, Newcastle-upon-Tyne Hospitals Foundation Trust, Newcastle-upon-Tyne, UK

²NIHR Newcastle In Vitro Diagnostics Co-operative, Translational and Clinical Research Institute, Newcastle University, Newcastle-upon-Tyne, UK

³Department of Surgery and Cancer, Faculty of Medicine, Imperial College London, UK

⁴Division of Infection, Immunity & Respiratory Medicine, University of Manchester, UK

⁵Humanitarian Conflict Response Institute, School of Arts, Languages and Cultures, University of Manchester, UK

⁶Division of Medical Sciences and Graduate Entry Medicine, University of Nottingham, UK

⁷NIHR Applied Research Collaboration East Midlands (ARC-EM), Nottingham, UK

⁸NIHR Community Healthcare MedTech and In Vitro Diagnostics Cooperative, Oxford Health NHS Foundation Trust, Oxford, UK

⁹Division of Health Sciences, University of Warwick, Coventry, UK

¹⁰Department of Geratology, Oxford University Hospitals NHS Foundation Trust, John Radcliffe Hospital, Oxford, UK ¹¹Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK

¹²NIHR Doctoral Research Fellowship Programme, UK

¹³Division of Cardiovascular Sciences, Faculty of Biology, Medicine and Health, University of Manchester, UK

¹⁴Healthcare Associated Infections Research Group, NIHR Leeds In Vitro Diagnostics Co-operative, Leeds Teaching Hospitals NHS Trust and University of Leeds, Leeds, UK

¹⁵Yorkshire and Humber Academic Health Science Network, Wakefield, UK

¹⁶Patient Public Involvement (PPI) Member, Precision Antimicrobial Prescribing PPI Group, NIHR Community Healthcare MedTech and In Vitro Diagnostics Cooperative, Nuffield Department of Primary Care Health Sciences, University of Oxford ¹⁷School of Clinical and Experimental Sciences, Faculty of Medicine, University of Southampton, Southampton, UK

¹⁸Department of Infection, University Hospital Southampton NHS Foundation Trust, Southampton, UK

¹⁹NIHR Southampton Biomedical Research Centre, University Hospital Southampton NHS Foundation Trust, Southampton, UK

²⁰Department of Respiratory Medicine, Kettering General Hospital NHS Foundation Trust, Kettering, UK

²¹Emergency Department, Manchester Royal Infirmary, Manchester University NHS Foundation Trust, UK

Corresponding author:

Dr. Samuel G. Urwin NIHR Newcastle In Vitro Diagnostics Co-operative, Medical School, Newcastle University, Framlington Place, NE2 4HH, Newcastle-upon-Tyne, UK. Email: <u>sam.urwin@newcastle.ac.uk</u>

NOTE: This preprint reports new research that has not been certified by peer review and should not be used to guide clinical practice.

Abstract

Background: Point-of-care (POC) tests for COVID-19 could relieve pressure on isolation resource, support infection prevention and control, and help commence more timely and appropriate treatment. We aimed to undertake a systematic review and pooled diagnostic test accuracy study of available individual patient data (IPD) to evaluate the diagnostic accuracy of a commercial POC test (FebriDx) in patients with suspected COVID-19.

Methods: A literature search was performed on the 1st of October 2020 to identify studies reporting diagnostic accuracy statistics of the FebriDx POC test versus real time reverse transcriptase polymerase chain reaction (RT-PCR) testing for SARS-CoV-2. Studies were screened for risk of bias. IPD were sought from studies meeting the inclusion and exclusion criteria. Logistic regression was performed to investigate the study effect on the outcome of the RT-PCR test result in order to determine whether it was appropriate to pool results. Diagnostic accuracy statistics were calculated with 95% confidence intervals (CIs).

Results: 15 studies were screened, and we included two published studies with 527 hospitalised patients. 523 patients had valid FebriDx results for Myxovirus resistance protein A (MxA), an antiviral host response protein. The FebriDx test produced a pooled sensitivity of 0.920 (95% CI: 0.875-0.950) and specificity of 0.862 (0.819-0.896) compared with RT-PCR, where there was an estimated true COVID-19 prevalence of 0.405 (0.364-0.448) and overall FebriDx test yield was 99.2%. Patients were tested at a median of 4 days [interquartile range: 2:9] after symptom onset. No differences were found in a sub-group analysis of time tested since the onset of symptoms.

Conclusions: Based on a large sample of patients from two studies during the first wave of the SARS-CoV-2 pandemic, the FebriDx POC test had reasonable diagnostic accuracy in a hospital setting with high COVID-19 prevalence, out of influenza season. More research is required to determine how FebriDx would perform in other healthcare settings with higher or lower COVID-19 prevalence, different patient populations, or when other respiratory infections are in circulation.

Trial registration: This work was based on a pooled analysis of anonymised data from two previous studies; the CoV-19POC study, described by Clark et al. (9), the "Southampton study" [ISRCTN:14966673, date registered: 18/03/2020]; and a study described by Karim et al. (13) the "Kettering study".

Keywords: FebriDx, MxA, COVID-19, point of care, diagnostic accuracy

Lay summary

Tests to diagnose COVID-19 are crucial to help control the spread of the disease and to guide treatment. Over the last few months, tests have been developed that can detect the SARS-CoV-2 virus which causes COVID-19. These tests use complex machines in pathology laboratories accepting samples from large geographical areas. Sometimes it takes days for test results to come back. So, to reduce the wait for results, new portable tests are being developed. These point-of-care (POC) tests are designed to work close to where patients require assessment and care such as hospital emergency departments, GP surgeries or care homes. For these new POC tests to be useful, they should ideally be as good as standard laboratory tests so patients get their result quickly and can benefit from the best, safest care.

In this study we looked at published research into a new test, FebriDx, which can detect the presence of any viral infection, including infections due to the SARS-CoV-2 virus, as well as bacterial infections which can have similar symptoms. The FebriDx result was compared with that obtained on the same patient's throat and nose swab and using the standard COVID-19 viral laboratory test. We were able to analyse data from two studies with a total of 523 adult patients who were receiving emergency hospital care with symptoms of COVID-19 during the early stage of the UK pandemic. Almost half of the patients were diagnosed as positive for SARS-CoV-2 virus using standard laboratory COVID-19 viral tests.

Our analysis demonstrated that the FebriDx POC test agreed 94 out of 100 times with the standard laboratory test results when FebriDx diagnosed the patient as free from COVID-19. However, FebriDx agreed only 82 out of 100 times with the standard laboratory test when FebriDx indicated that the patient had a COVID-19 infection. These differences have important implications for how these tests could be used. As there were far fewer FebriDx false results when the results of the FebriDx test were negative (6 out of 100) than when the results of the FebriDx test were positive (18 out of 100), we can have more confidence in a negative test result using FebriDx at the POC than a positive FebriDx result.

Overall, we have shown that the FebriDx POC test performed quite well during the first wave of the COVID-19 pandemic when compared with laboratory tests, especially when the POC test returned a negative test. For the future, this means that the FebriDx POC test might be helpful in making a rapid clinical decision whether to isolate a patient with COVID-19-like symptoms arriving in a busy emergency department. However, our results indicate it would not completely replace the need to conduct a confirmatory laboratory test in certain cases.

There are limitations to our findings. For example, we do not know if FebriDx will work in a similar way with patients in different settings such as in the community or care homes. Similarly, we do not know whether other viral and bacterial infections which cause similar COVID-19 symptoms, and are more common in the autumn and winter months, could influence the FebriDx test accuracy.

Background

The global SARS-CoV-2 pandemic (1) has put considerable pressure on healthcare services worldwide. Health and care providers require diagnostic strategies to rapidly identify patients infected with SARS-CoV-2 to implement the accurate segregation of positive and negative patients in health and care facilities, and to ensure early administration of evidence-based therapies to patients with coronavirus disease 2019 (COVID-19) (2). The risks of nosocomial infection are high (3) and mechanisms to ensure that SARS-CoV-2 has limited transmission within hospitals, care facilities, and the community are an urgent priority as many countries begin to face a second wave of infection.

There has been rapid development of novel clinical tests to support screening and diagnosis in both symptomatic and asymptomatic patients. In particular, there have been a number of molecular and antibody tests which manufacturers have developed for use at the point-of-care (POC) (4). POC tests for influenza have been adopted in many hospitals and other health and care facilities (5), where the rapid availability of results inform patient management, selection of the appropriate location for ongoing healthcare provision, and infection prevention and control (6, 7). In the emergency department setting for example, a rapid COVID-19 test result to aid triage of the patient into the appropriate COVID-19/non-COVID-19 sections of the hospital may contribute to a reduction in nosocomial infection, providing significant benefits to patient pathways, workflows, and outcomes (8).

There is limited published evidence on the accuracy, reliability, and usability of many of the available POC tests for COVID-19, in particular, when used in-context within clinical settings with varying disease prevalence. Some pre-existing POC tests may have a role in the management of patients with suspected COVID-19. The FebriDx lateral flow device (LFD) (Lumos Diagnostics, Sarasota, Florida, USA) is a CE marked POC test that detects two host response proteins, myxovirus resistance protein A (MxA) and C reactive protein (CRP), in finger prick blood samples. This combination of MxA and CRP is designed to distinguish between viral and bacterial respiratory infection (9-11). MxA is an intracellular protein that is exclusively induced by type I interferon (IFN) and not by other cytokines expressed during bacterial infection (12, 13). Type I IFNs are produced in response to a wide range of viral infections and are found to be elevated in the presence of most acute viral infections (10), therefore providing strong theoretical grounds to expect MxA to rise in response to SARS-Cov-2 infection. The manufacturer's intended use includes recommendations to use in patients older than 2 years presenting within 3 days of an acute onset fever (exhibited or reported) and within 7 days of new onset respiratory symptoms consistent with a community-acquired upper respiratory infection (14).

We undertook a systematic review and pooled diagnostic test accuracy study of available individual patient data (IPD) to evaluate the diagnostic accuracy of the FebriDx LFD compared to contemporaneous reverse transcriptase polymerase chain reaction (RT-PCR) testing to understand the performance of FebriDx in the identification of patients with COVID-19. We did not limit this analysis to patients presenting within a certain number of days since the onset of symptoms in order to inform and identify all potential use cases within the pandemic.

Method

We performed a systematic review with a pooled diagnostic test accuracy study of available IPD, and followed the STAndards for Reporting Diagnostic accuracy studies (STARD) checklist (15) (supplementary material 1). Although not formally a systematic review and meta-analysis of individual patient data (IPD) (16), we also followed the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA)-IPD checklist (PRISMA-IPD) (17) (supplementary material 2), where applicable. This study was conducted at pace as part of the CONDOR national test evaluation programme (18), and as a result, no protocol was developed, and the study was not registered.

Literature search

The inclusion criteria for studies were: published or un-published (preprint) diagnostic test accuracy studies; FebriDx used as the index test; RT-PCR used as a comparator test; and an adult population suspected of COVID-19 regardless of the time since symptom onset. The exclusion criteria were studies that were not a diagnostic test accuracy study.

OVID Medline, LOVE Platform (Epistemonikos, Santiago, Chile), and the references from the Living Systematic Review on SARS-CoV-2 (19) were electronically searched on the 30th of July 2020, and again on the 1st of October 2020, by one author (JS) to identify diagnostic test accuracy studies reporting diagnostic accuracy statistics of FebriDx versus RT-PCR for SARS-CoV-2. The following keywords: "COVID-19", "2019-nCOV", "SARS-COV-2", "novel coronavirus disease" AND "FebriDx" were used. No date, location or language restrictions were applied to the search results, and all of the databases included pre-prints.

The abstracts of the search results were accessed and screened against the inclusion and exclusion criteria by two authors independently (SGU and KG). The two authors discussed, compared, and combined their findings, and if there was disagreement, adjudication was provided by a third author (AJA). If there was insufficient information in the abstract to exclude the study based on the inclusion and exclusion criteria, then a conservative approach of accessing the full text to perform the screening was taken to mitigate the risk of erroneously excluding relevant studies.

Risk of bias (RoB) assessments were performed on the manuscripts of the identified studies that passed eligibility screening against the inclusion and exclusion criteria using the QUADAS-2 tool (20) for the quality assessment of diagnostic accuracy studies by two authors independently (SGU and KG). The QUADAS-2 binary prompts were deemed insufficient to fully capture the potential bias of the studies. The descriptive sections of the tool were therefore used to capture any other potential sources of bias not considered by the binary prompts. These additional aspects contributed to the final assessment for RoB. The two authors held a discussion to compare and combine their findings, and if there was disagreement, this was adjudicated by a third author (BCL).

The Chief Investigators (CIs) of the identified studies that passed eligibility screening against the inclusion and exclusion criteria were approached via email to provide anonymised IPD. If IPD were not available, or not provided, then the study was excluded from further analysis. The minimum data

set that was requested from the CIs is outlined in supplementary material 3. Any queries relating to the study or provided data were communicated and resolved via email. The RoB assessments were performed prior to making contact with the CIs, and any additional information provided by the CIs not included in the manuscripts did not impact upon the RoB assessments.

Data processing

Following receipt of the IPD, the FebriDx and RT-PCR results were summarised in 2x2 contingency tables for each study independently and also pooled across all studies. The following assumptions were applied. For FebriDx: a viral result (i.e. MxA positive [≥40 ng/ml], regardless of CRP positivity) was considered positive; a non-viral result (i.e. MxA negative [<40 ng/ml], CRP positive [≥20mg/L]) was considered negative; and a negative result (i.e. MxA negative [<40 ng/ml], CRP negative [<20mg/L]) was considered negative. For RT-PCR: if COVID-19 was detected the result was considered positive (COVID-19 + 'other' were also considered positive if studies used a respiratory panel that also tested for other viruses); and if COVID-19 was not detected the result was considered negative.

Data analysis

A complete case analysis approach was taken whereby cases with a completed and valid FebriDx and RT-PCR result pair were included, regardless of missing data within other fields. This was undertaken to maximise the available sample size for the pooled analysis. Cases that had missing FebriDx or RT-PCR results were therefore excluded, in addition to cases that had a final indeterminate FebriDx or RT-PCR result for the pooled analysis. However, to estimate overall test yield to allow interpretation of diagnostic accuracy estimates in spite of this, we included all cases with missing or invalid FebriDx test results.

To determine if the study populations were similar, and to quantify any baseline and outcome imbalance, the distributions of variables were compared statistically between the included studies. Non-parametric two-sample Wilcoxon rank sum tests (two-sided) were used to compare numerical data, whilst two-sample tests for equality of proportions with continuity corrections were used to compare categorical data. Statistical significance was set at an alpha level of <0.05. Numerical data were summarised using the median and interquartile range, whilst categorical data were summarised using counts and proportions.

To formally determine whether it was appropriate to pool the diagnostic accuracy results from the included studies, logistic regression was undertaken to investigate the study effect on the outcome of the RT-PCR test result, whilst controlling for the FebriDx test result. Thus a main effect for study, and an interaction effect between the FebriDx test result and study were included in the model. The following logistic regression model was constructed using a binomial error distribution and logit link function:

PCR test result = *FebriDx* test result + *Study* + (*FebriDx* test result * *Study*)

Diagnostic accuracy measures with 95% confidence intervals (CIs) were calculated from the 2x2 contingency tables for each study independently and also pooled across studies (supplementary material 4). A sub-group analysis was also performed, with the pooled data stratified by the time tested since the onset of symptoms into two groups: 0 to 7 days, and >7 days. This was undertaken to determine whether the time tested since the onset of symptoms had an impact on the diagnostic accuracy of FebriDx. The following diagnostic accuracy measures were calculated across all study groups:

- Test yield (the proportion of all patients tested with FebriDx that had a valid test result from FebriDx)
- Apparent prevalence (the proportion of all patients with COVID-19 according to FebriDx)
- Estimated true prevalence (proportion of all patients with COVID-19 according to RT-PCR)
- Test sensitivity (proportion of patients with COVID-19, according to RT-PCR, that test positive by FebriDx)
- Test specificity (proportion of patients without COVID-19, according to RT-PCR, that test negative by FebriDx)
- Diagnostic accuracy (proportion of all FebriDx tests that give a correct result according to RT-PCR)
- Positive predictive value (PPV) (probability that patients with a positive FebriDx test have COVID-19 according to RT-PCR)
- Negative predictive value (NPV) (probability that patients with a negative FebriDx test do not have the disease according to RT-PCR)
- Positive likelihood ratio (probability of a patient who has COVID-19 according to PCR testing positive by FebriDx, divided by the probability of a patient who does not have the disease according to RT-PCR testing positive by FebriDx)
- Negative likelihood ratio (probability of a patient who has COVID-19 according to RT-PCR testing negative by FebriDx, divided by the probability of a patient who does not have the disease according to RT-PCR testing negative by FebriDx)

Results

Fifteen studies were identified from the literature search, with seven studies excluded following deduplication. Six studies were excluded following screening against the inclusion and exclusion criteria, leaving two studies for potential inclusion (supplementary material 5). The CIs of the two studies were able to provide IPD which were included in analysis. The two studies were from the UK: a study from Southampton (CI = TWC) (9) and a study from Kettering (CI = RVR) (13). In addition to receiving the requested data underpinning the publications, an additional, unpublished dataset was made available to us from the Southampton study using the same methods as their initial publication (9).

Risk of bias

The results of the RoB assessment are provided in supplementary material 6. In the 'Patient selection' domain, the Southampton study was assessed to have an unclear RoB due to uncertainty around which patients were re-tested and which patients were excluded, and whether those with no CRP line [<20mg/L] on the FebriDx LFD but with concurrent laboratory CRP results of ≥20mg/L were excluded. In the 'Index test' domain, the Kettering study was assessed to have an unclear RoB due to patients with FebriDx positive and RT-PCR negative tests considered to be positive for COVID-19 if clinical suspicion was high. Due to this, there was concern regarding the applicability of the index test to the focus of this study. In the 'Flow and timing domain', the Kettering study was assessed to have a high risk of RoB due to the exclusion of one third of the patients for having symptoms longer than 7 days, however this did follow FebriDx's instructions for use (14).

Eligibility criteria, index test, and reference standard in the included studies

Table 1 presents a summary of the eligibility criteria, index test, and reference standard used within the Southampton and Kettering studies.

	Southampton (9)	Kettering (13)			
Inclusion	 Age ≥18 years old Patient in ED, AMU, HDU, GICU, medical wards, or another location within Southampton General Hospital, University Hospital Southampton NHS Foundation Trust (UHS) Can be recruited to the study within 24 hours of presentation to hospital AND Has acute respiratory illness (ARI)⁺ OR Does not have ARI but is a suspected case of COVID-19 according to the current PHE case definition OR Does not have ARI or fulfil the PHE case definition of a suspected case but testing for SARS-COV-2 is considered necessary by the responsible clinical team 	 Age >16 years old Require admission for at least one night with suspected COVID-19 infection Meet the PHE criteria for swab testing for COVID-19 which includes the requirement of hospital admission and having: Clinical or Radiological evidence of pneumonia or; Acute respiratory distress syndrome or; Influenza like illness (fever ≥37.8°C and at least one of the following respiratory symptoms, which must be of acute onset (within 7 days of enrolment): persistent cough (with or without sputum), hoarseness, nasal discharge or congestion, shortness of breath, sore throat, wheezing, sneezing)), or; Inpatients with new respiratory symptoms or fever without another cause or worsening of a pre-existing respiratory condition 			
Exclusion	 Not fulfilling all the inclusion criteria Declines nasal / pharyngeal swabbing Consent declined or consultee consent declined Already recruited to the study in the last 14 days 	 Unable/unwilling to consent to the study Does not meet PHE criteria for swab testing Taking antibiotics or antivirals in the last 14 days Received Live vaccine in the last 14 days 			
Index test	• FebriDx	• FebriDx			
Reference	 QIAstat-Dx polymerase chain reaction (PCR) system for SARS-CoV-2 testing PHE laboratory RdRp and envelope protein (E) gene PCR assays 	 PHE laboratory RdRp and envelope protein (E) PCR assays‡ 			

Table 1 - The eligibility criteria, index test, and reference standard of the included studies

'ARI' = 'Acute respiratory illness'; 'ED' = 'Emergency Department'; 'AMU' = 'Acute Medical Unit'; 'HDU' = 'High Dependency Unit'; 'GICU' = 'General Intensive Care Unit'; 'PHE' = 'Public Health England'; 'RdRp' = 'RNA-dependent RNA polymerase' †An episode of acute respiratory illness is defined as an acute upper or lower respiratory illness (including rhinitis, rhinosinusitis, pharyngitis, pneumonia, bronchitis and influenza-like illness) or an acute exacerbation of a chronic respiratory illness (including exacerbation of chronic obstructive pulmonary disease, asthma or bronchiectasis). For the study, acute respiratory illness as a provisional, working, differential or confirmed diagnosis must be made by a treating clinician; ‡The European Centre for Disease Prevention and Control COVID-19 Case Definition was also used as a reference standard, but that is not relevant to this work, and is therefore not presented here.

Eligibility criteria

In the Southampton study, consecutive patients were approached for participation at Southampton General Hospital, University Hospital Southampton NHS Foundation Trust, between the 20th March 2020 and 29th April 2020. In the Kettering study, consecutive patients were approached for participation at Kettering General Hospital, Kettering General Hospital NHS Foundation Trust, between 16th of March and 7th of April 2020.

When combined, both studies recruited hospitalised patients over 16 years of age with suspected COVID-19 between the 16th of March and 29th April 2020, however the Southampton study also allowed patients to be recruited who did not have an acute respiratory illness (ARI) or did not meet

the PHE definition of a suspected case, but where testing was considered necessary by the clinical team.

Index test

In both studies, the results of the FebriDx test were not shared with the clinical teams and the readers of the FebriDx test lines were blinded to the RT-PCR results, and vice versa (9). In both studies, the finger prick blood samples for FebriDx were taken at the same time as the nose and throat swabs for RT-PCR (9, 13). In the Southampton study, the FebriDx result was read independently by two investigators and disagreements were further adjudicated by a third investigator. In the Kettering study, the FebriDx result was read by one investigator, however if the result was inconclusive or negative, this was further adjudicated by two investigators.

Reference standard

The Southampton study used the QIAstat-Dx RT-PCR system (Qiagen, Hilden, Germany) for analysing combined nose and throat swabs, which gave a binary readout of positive or negative for the detection of targets including SARS-CoV-2 (9), in addition to Public Health England (PHE) laboratory RNA-dependent RNA polymerase (RdRp) and envelope protein (E) RT-PCR testing for SARS-CoV-2. However, only results from the QIAstat-Dx RT-PCR system were available for the additional unpublished data from the Southampton study following their initial publication (9), so the results from the QIAstat-Dx RT-PCR system were used for all patients from the Southampton study to maintain within-study consistency for the purposes of this analysis. The Kettering study used Public Health England laboratory RdRp and envelope protein (E) RT-PCR testing for SARS-CoV-2 to analyse nose and throat swabs (13).

Patient flow

The flow of patients in the Southampton and Kettering studies are presented in supplementary material 7 and 8, respectively, and pooled data from both studies is presented in Figure 1.





Figure 1 - Flow diagram of the pooled data from the Southampton (Clark et al. (9) + additional unpublished data) and Kettering studies (Karim et al. (13))

In the Southampton study, 500 patients were considered for testing with FebriDx, with 22 excluded (4.4%) as it was deemed inappropriate by the clinical team, or where the patient/carer declined participation in the study. Out of the 478 patients tested with FebriDx, 19 tests were initially invalid

(4%). FebriDx could not be repeated in 3 of the 19 initially invalid tested patients (15.8%). Out of the 16 initially invalid tested patients that were retested, 1 was invalid (6.3%), and was subsequently retested again where the patient then received a valid test result upon a second retest. Considering all 20 invalid tests, 16 were due to blood clotting in the collection tube (80%), whilst 4 were due to there being no CRP line [<20mg/L] on the FebriDx device but with concurrent laboratory CRP results of \geq 20mg/L (20%). 475 patients remained for analysis, resulting in a test yield of 99.4% (supplementary material 7). The Kettering study approached 75 patients for testing with FebriDx, where 26 were excluded (34.7%), with 25 due to symptoms being longer than 7 days, and 1 due to being immunosuppressed. Out of the 49 patients tested with FebriDx, 1 test was initially invalid (2%) due to the inability to obtain enough blood. FebriDx could not be repeated in this patient as they were elderly, frail, and clinically unstable at the time of testing, and 48 patients remained for analysis, resulting in a test yield of 98% (supplementary material 8).

This resulted in 575 patients being approached for testing with FebriDx when pooled, with 48 excluded due to the aforementioned reasons (8.4%). Out of the 527 patients tested with FebriDx, 20 tests were initially invalid (3.8%). FebriDx could not be repeated in 4 of the 20 initially invalid tested patients (20%). Out of the 16 initially invalid tested patients that were retested, 1 was invalid (6.3%), and was subsequently retested again where the patient then received a valid test result upon a second retest. 523 patients remained for the pooled analysis, resulting in an overall test yield of 99.2%.

Patient population and outcome

The patient population and outcome in the Southampton study, Kettering study, and pooled data from both studies is summarised in Table 2.

	Southampton (9)		Kettering (13)		Pooled		Comp
	n/med	%/IQR	n/med	%/IQR	n/med	%/IQR	p value
Population							
Valid tests (n)	475		48		523		
Age (median, IQR)	69	52-81	67	54.3-78	69	52-81	0.59
Male (n, %)	244	51.4	32	66.7	276	52.8	0.06
Symptom duration (med, IQR) ⁺	5	1-10	3	2:5	4	2:9	0.15
Symptoms‡:							
Sore throat (n %)	92	24.7	0	0.0			
Rhinorrhoea (n, %)	91	24.4	0	0.0			
Wheeze (n, %)	135	35.5	0	0.0			
SOB (n, %)	297	71.9	38	79.2			
Pleuritic pain (n, %)	96	25.3	2	4.2			
Cough (n, %)	240	59.0	40	83.3			
Sputum (n <i>,</i> %)	105	27.8	0	0.0			
Fever (n, %)	192	48.2	41	85.4			
Chills (n, %)	165	43.4	0	0.0			
Fatigue (n, %)	255	68.0	6	12.5			
Reduced appetite (n, %)	224	60.2	2	4.2			
Headache (n, %)	148	39.8	0	0.0			
Myalgia (n, %)	115	31.1	3	6.2			
Diarrhoea (n, %)	97	25.5	3	6.2			
Abdominal pain (n, %)	60	16.0	3	6.2			
Anosmia (n, %)	66	18.8	0	0.0			
Outcome							
Death‡	74	17.8	12	25.0			

Table 2 - Patient population and outcome in the included studies and pooled data from both studies

'Comp' = 'Comparison'; 'n' = 'number'; 'med' = 'median'; 'IQR' = 'interquartile range'; 'SOB' = 'shortness of breath' †Symptom duration is the number of days from symptom onset until the samples were obtained for testing by FebriDx/RT-PCR (samples were collected for FebriDx and RT-PCR at the same time)

‡Statistical analysis not performed between studies due to methodological differences in how these data were defined and captured

Symptom and outcome data were provided as requested for both studies, but due to methodological differences between the studies in how these data were defined and captured, they were excluded from statistical analyses. The Southampton study recorded symptoms in the prospectively stipulated categories presented in Table 2, whereas the Kettering study recorded free text symptoms. For presentational purposes only, two authors (SGU, AJA) reviewed each of the free text fields in the Kettering study independently and applied the same categories as the Southampton study, with no disagreements requiring adjudication by a third author. The Southampton study reported death at 30 days following admission, whereas the Kettering study reported death at the end of the index admission.

No statistically significant differences were found between the populations of the Kettering and Southampton studies in terms of age (p = 0.59), sex (p = 0.06), and symptom duration (p = 0.15), although the sex distributions almost reached statistical significance, with the Kettering study having a slightly higher proportion of males (66.7% vs. 51.4%).

Diagnostic test accuracy

Pooling data

In the logistic regression analysis to investigate the study effect on the outcome of the RT-PCR test result, none of the included terms were statistically significant: FebriDx test result (b = 18.6; p = 0.98); the study (b = 13.9; p = 0.98); and an interaction term between FebriDx test result and the study (b = -14.5; p = 0.98). The interaction term was therefore removed from the hierarchical model, and in the reduced model, as expected, the FebriDx test result was highly statistically significant (b = 4.24; p<0.001), and the study effect was still not statistically significant (b = -0.32; p=0.49). This result suggests little study effect on the outcome of the PCR test result, whilst controlling for the FebriDx test result, and supports pooling of data across the studies.

Results

The Southampton and Kettering studies had a test yield of 99.4% and 98%, respectively, and when combined, a test yield of 99.2% was found. These valid tests were presented in 2x2 contingency tables for the Southampton study, Kettering study, and pooled data (including sub-group analyses) from both studies in Figure 2, with the diagnostic accuracy results calculated for the Southampton study, Kettering sub-group analyses) from both studies presented in Table 3.

Southampton (9)

n = 475	RT-PCR +	RT-PCR -	Total		
FebriDx +	164	39	203		
FebriDx -	17	255	272		
Total	181	294	475		

Kettering (13)

n = 48	RT-PCR +	RT-PCR -	Total
FebriDx +	31	4	35
FebriDx -	0	13	13
Total	31	17	48

Pooled [all]

n = 523	RT-PCR +	RT-PCR -	Total
FebriDx +	195	43	238
FebriDx -	17	268	285
Total	212	311	523

Pooled [sub-group]: symptoms 0 to 7 days

n = 372	RT-PCR +	RT-PCR -	Total
FebriDx +	136	29	165
FebriDx -	10	197	207
Total	146	226	372

Pooled [sub-group]: symptoms >7 days

n = 150	RT-PCR +	RT-PCR -	Total
FebriDx +	59	14	73
FebriDx -	7	70	77
Total	66	84	150

Figure 2 - 2x2 contingency tables from the Southampton study (Clark et al. (9) + additional unpublished data), Kettering study (Karim et al. (13)), and pooled [all and sub-group] data from both studies. One patient was excluded from the 'Pooled [sub-group]' analysis due to missing symptom duration data. Please note that data from the 'Pooled [sub-group]: symptoms >7 days' were from the Southampton study only

	Southampton (9)		Kettering (13)		Pooled [all]		Pooled [sub-group]: symptoms 0-7 days‡		Pooled [sub-group]: symptoms >7 days‡¥	
	n = 475		n = 48		n	n = 523		= 372	n = 150	
	Estimate	95% CI	Estimate	95% CI	Estimate	95% CI	Estimate	95% CI	Estimate	95% CI
Apparent prevalence	0.427	0.384-0.472	0.729	0.589-0.835	0.455	0.413-0.498	0.444	0.394-0.494	0.487	0.408-0.566
True prevalence ⁺	0.381	0.338-0.426	0.646	0.504-0.766	0.405	0.364-0.448	0.392	0.344-0.443	0.440	0.363-0.520
Sensitivity	0.906	0.854-0.941	1.000	0.869-1.000	0.920	0.875-0.950	0.932	0.877-0.964	0.894	0.794-0.951
Specificity	0.867	0.824-0.902	0.765	0.522-0.910	0.862	0.819-0.896	0.872	0.821-0.910	0.833	0.738-0.899
Accuracy	0.882	0.85-0.908	0.917	0.799-0.972	0.885	0.855-0.910	0.895	0.860-0.923	0.860	0.795-0.907
PPV	0.808	0.748-0.856	0.886	0.735-0.961	0.819	0.765-0.863	0.824	0.759-0.875	0.808	0.702-0.883
NPV	0.938	0.902-0.961	1.000	0.734-1.000	0.940	0.906-0.963	0.952	0.912-0.975	0.909	0.821-0.958
PLR	12.926	8.545-22.476	Inf	4.473-Inf	13.736	9.087-23.993	17.062	9.978-36.295	8.890	4.886-23.923
NLR	0.205	0.149-0.266	0.114	0.028-0.255	0.192	0.142-0.247	0.185	0.126-0.249	0.211	0.117-0.319

Table 3 – Diagnostic accuracy results for the included studies and pooled data from both studies

'Cl' = 'confidence interval'; 'PPV' = 'positive predictive value'; 'NPV' = 'negative predictive value'; 'PLR' = 'positive likelihood ratio'; 'NLR' = 'negative likelihood ratio'; 'Inf' = 'infinity';

†Estimated true prevalence (i.e. proportion of all patients with COVID-19 according to RT-PCR)

‡One patient was excluded from the 'Pooled [sub-group]' analysis due to missing symptom duration data

¥These data were from the Southampton study only

The estimated true and apparent COVID-19 prevalence were the only statistically significant differences between the studies in the diagnostic accuracy measures (both p<0.001), with the Kettering study experiencing a higher COVID-19 prevalence (estimated true prevalence = 0.646 (95% CI: 0.504-0.766)) than the Southampton study (estimated true prevalence = 0.381 (0.338-0.426)).

The pooled results from 523 patients gave an estimated sensitivity of 0.920 (0.875-0.950) and specificity of 0.862 (0.819-0.896) for the FebriDx test, with an overall diagnostic accuracy of 0.885 (0.855-0.910) at an estimated true prevalence of 0.405 (0.364-0.448) (Table 3). The PPV was 0.819 (0.765-0.863) and NPV was 0.940 (0.906-0.963).

One patient was excluded from the sub-group analyses of the pooled results due to missing symptom duration data. In these analyses, no statistically significant differences were apparent between the 372 patients that were tested between 0 and 7 days after symptom onset and the 150 patients that were tested more than 7 days after symptom onset in any of the diagnostic accuracy measures calculated (Table 3).

Discussion

In this systematic review and pooled analysis of IPD, we found that the FebriDx LFD had a pooled sensitivity of 0.920 (95% CI: 0.875-0.950) and specificity of 0.862 (0.819-0.896) for COVID-19 across two studies performed within acute hospitals in the UK when compared to RT-PCR on nose and throat swabs during the first wave of the SARS-CoV-2 pandemic. There were no other published data on the diagnostic accuracy of FebriDx, with the overall evidence base in terms of studies performed remaining small. The two studies did, however, include a total of 523 patients, achieving a large sample size for the pooled analysis.

Testing with FebriDx produced an initial test failure rate of 3.8% (20 out of 527 patients tested). In those patients where re-testing was performed, most were valid upon the first re-test (15 out of 16 patients), and only one patient required a second re-test to produce a valid result. Most test failures were described as being the result of blood clotting within the collection tube, and therefore not being released into the device for analysis.

The duration between onset of symptoms and the patient being tested did not seem to have an impact on the diagnostic accuracy of FebriDx, with no statistically significant differences evident between the two sub-groups of 0 to 7 days and greater than 7 days after symptom onset. However, only one of the studies included patients with symptom onset after 7 days, so this finding is not from a pooled analysis.

All patients included in the pooled analysis were from the acute hospital setting, and such findings must be extrapolated with caution to other patient groups and settings. Further context-specific evaluation would be required in order for FebriDx to be used in other patient groups where performance has not yet been demonstrated; such as children, immunocompromised and cancer patients, and those who are asymptomatic or pauci-symptomatic. Further, taking care homes as an example setting, the mean age of residents is 85 years (21), and all care home residents are significantly affected by frailty. The high prevalence of immunosenescence in this group is such that CRP and MxA results might be significantly attenuated (22). It is clear that hospital data cannot be extrapolated to such a group and further context-specific evaluation would be required in other settings.

In the context of older, more frail, community dwelling populations where delirium is a common, sensitive, but non-specific presentation of COVID-19 (23), the ability to rule out viral and bacterial infections as the cause of delirium during an outbreak may be even more important than the ability to detect them. Further work is therefore required to look at the NPV of FebriDx in populations where such information may be of use.

It should also be noted that specific treatments (neuraminidase inhibitors) are available and recommended for use during influenza outbreaks, and an increasing body of evidence supports different interventions (remdesivir and dexamethasone) for use in cases of COVID-19 (2, 24). The ability to distinguish between different types of viral infection is important and the value added by FebriDx in the context of already wide availability of influenza POC testing is unclear.

Although not a direct test for the presence of SARS-CoV-2 infection, the performance measures reported in this analysis are comparable to results from other studies of FebriDx in detecting the presence of respiratory viral infections (10). As a raised MxA level is not diagnostic of SARS-CoV-2 due to its non-specific response to a number of respiratory infections, the optimal use case of the FebriDx test is unlikely to be for 'ruling in' COVID-19. In a hospital setting, if FebriDx was used to cohort patients to wards incorrectly, then the unspecific nature of the result may lead to the exposure of patients to potentially serious co-infection. Recent evidence has suggested that the risk of death from co-infection of SARS-CoV-2 and influenza was nearly double that of SARS-CoV-2 alone, 43.1% vs 26.9% (25). However, the simplicity of a fingerprick blood test with a 10-minute turnaround time could enable rapid 'rule out' of COVID-19 in patients who have low concentrations of MxA. In a hospital setting, those patients could be potentially sent to non-COVID areas of a hospital.

In the pooled analysis presented in this study, the NPV was 0.940 at a prevalence of 0.405. The performance of FebriDx as a rule out test at a lower prevalence is likely to be better. In the community, the test could potentially be used to allow relatives to visit care homes residents; to facilitate air travel; or to support schools or universities. However, it is imperative to evaluate this test within the appropriate patient populations in-context for each of these use cases before making recommendations. In this study, we have only identified evidence relating to use in symptomatic patients presenting to hospital with suspected COVID-19.

Implications for practice

Whilst diagnostic performance measures were reasonably high, our findings are limited by the uncertain generalisability to longer durations of symptoms (more than a week), to different healthcare settings, and to different phases of the pandemic as prevalence rates will vary.

The utility of FebriDx may be limited to its ability to rule out acute COVID-19 infection because a positive result does not specify which respiratory viral pathogen is present. However, a sensitivity of 0.920 will lead to a false negative result in almost one in ten positive patients. This is an important consideration for settings with a high prevalence of disease in tested patients. Testing protocols would need to be developed carefully to ensure the correct use of the test, likely as a triage test in conjunction with RT-PCR.

If further evidence confirms the diagnostic characteristics of FebriDx in a hospital setting then it may be of greatest utility when deployed in a triaging capacity. For example, enabling the allocation of patients to wards based on the likely risk of SARS-CoV-2 whilst confirmatory RT-PCR testing is sought. This should, however, be used with caution due to the potential increased risk that coinfection poses to patients with SARS-CoV-2. The use of the FebriDx LFD should be carefully considered within the context of both clinical pathway needs and the patient pathways that it may influence.

Limitations

The reference standard of RT-PCR on nose and throat swab samples is imperfect, and while commonly used as a reference test, it is not a gold standard (26). RT-PCR has shown limited diagnostic performance characteristics, particularly with the production of false negative results in patients presenting in an emergency with suspected COVID-19 (27-29). An imperfect reference standard in this case, which most likely produced false negative results, would be likely to produce an underestimate of both sensitivity and specificity. If additional clinical and diagnostic data were available for both studies, this analysis would have benefitted from the use of a composite reference standard (30) or latent class analyses with instrumental variables to minimise the probability of such error or bias (31-33). Although both studies used RT-PCR as the reference standard, different RT-PCR tests were used across the studies. The accuracy of the different RT-PCR tests used will vary between methods, and as these methods were further developed during the pandemic, this leads to an inconsistent reference standard. This suggests that the pooled diagnostic accuracy estimates should be treated with caution.

Both studies were conducted during the first peak of the SARS-CoV-2 pandemic in the UK (March/April 2020), where the prevalence of COVID-19 was high. The pooled diagnostic accuracy results (particularly PPV and NPV) should only be interpreted within this specific phase of the pandemic, and not extrapolated to other phases where a lower prevalence of COVID-19 was evident. A key additional issue will occur when there are increased rates of several other viruses and bacteria circulating, most notably during the winter season. The possibility of co-colonisation/infection at such times will be a challenge for diagnostic accuracy evaluations. In particular, it is likely that the capacity to rule out COVID-19 will be compromised when other respiratory viruses are prevalent.

Conclusion

Based on a large sample of patients from two studies during the first wave of the SARS-CoV-2 pandemic, FebriDx had reasonable diagnostic accuracy in a hospital setting with high COVID-19 prevalence, out of influenza season. We cannot be certain how FebriDx would perform in other healthcare settings with higher or lower COVID-19 prevalence or at times year when other respiratory infections may affect diagnostic performance. Further evidence is needed on FebriDx's diagnostic performance and utility in different populations, clinical and non-clinical settings.

References

1. Zhou F, Yu T, Du R, Fan G, Liu Y, Liu Z, et al. Clinical course and risk factors for mortality of adult inpatients with COVID-19 in Wuhan, China: a retrospective cohort study. Lancet. 2020;395(10229):1054-62.

Horby P, Lim WS, Emberson JR, Mafham M, Bell JL, Linsell L, et al. Dexamethasone in 2. Hospitalized Patients with Covid-19 - Preliminary Report. N Engl J Med. 2020.

Klompas M. Coronavirus Disease 2019 (COVID-19): Protecting Hospitals From the Invisible. 3. Ann Intern Med. 2020;172(9):619-20.

FIND. SARS-COV-2 diagnostic pipeline. 2020. https://www.finddx.org/covid-19/pipeline/. 4. Accessed 14 Oct 2020.

Lee JJ, Verbakel JY, Goyder CR, Ananthakumar T, Tan PS, Turner PJ, et al. The Clinical Utility 5. of Point-of-Care Tests for Influenza in Ambulatory Care: A Systematic Review and Meta-analysis. Clin Infect Dis. 2019;69(1):24-33.

6. Roy K, Kandil H, Knight M, Thapa M, Groom K. Respiratory viral point of care testing (POCT) allows improved infection control and bed management during an influenza. European Respiratory Journal. 2018:52.

Youngs J, Marshall B, Farragher M, Whitney L, Glass S, Pope C, et al. Implementation of 7. influenza point-of-care testing and patient cohorting during a high-incidence season: a retrospective analysis of impact on infection prevention and control and clinical outcomes. J Hosp Infect. 2019;101(3):276-84.

8. Garcia-Castrillo L, Petrino R, Leach R, Dodt C, Behringer W, Khoury A, et al. European Society For Emergency Medicine position paper on emergency medical systems' response to COVID-19. Eur J Emerg Med. 2020;27(3):174-7.

Clark TW, Brendish NJ, Poole S, Naidu VV, Mansbridge C, Norton N, et al. Diagnostic accuracy 9. of the FebriDx host response point-of-care test in patients hospitalised with suspected COVID-19. J Infect. 2020.

10. Shapiro NI, Self WH, Rosen J, Sharp SC, Filbin MR, Hou PC, et al. A prospective, multi-centre US clinical trial to determine accuracy of FebriDx point-of-care testing for acute upper respiratory infections with and without a confirmed fever. Ann Med. 2018;50(5):420-9.

Self WH, Rosen J, Sharp SC, Filbin MR, Hou PC, Parekh AD, et al. Diagnostic Accuracy of 11. FebriDx: A Rapid Test to Detect Immune Responses to Viral and Bacterial Upper Respiratory Infections. J Clin Med. 2017;6(10).

12. Goodbourn S, Didcock L, Randall RE. Interferons: cell signalling, immune modulation, antiviral response and virus countermeasures. J Gen Virol. 2000;81(Pt 10):2341-64.

13. Karim N, Ashraf MZ, Naeem M, Anwar T, Aung H, Mallik S, et al. Utility of the FebriDx pointof-care test for rapid triage and identification of possible coronavirus disease 2019 (COVID-19). Int J Clin Pract. 2020:e13702.

14. FebriDx. Package Insert. 2015. https://www.rpsdetectors.com/wp-

content/uploads/2015/05/SPEC-MKT-044.2_FebriDx-Package-Insert-CE-Mark.pdf. Accessed 14 Oct 2020.

15. Bossuyt PM, Reitsma JB, Bruns DE, Gatsonis CA, Glasziou PP, Irwig LM, et al. Towards complete and accurate reporting of studies of diagnostic accuracy: the STARD initiative. BMJ. 2003;326(7379):41-4.

16. Riley RD, Lambert PC, Abo-Zaid G. Meta-analysis of individual participant data: rationale, conduct, and reporting. BMJ. 2010;340:c221.

Stewart LA, Clarke M, Rovers M, Riley RD, Simmonds M, Stewart G, et al. Preferred 17. Reporting Items for Systematic Review and Meta-Analyses of individual participant data: the PRISMA-IPD Statement. JAMA. 2015;313(16):1657-65.

COVID-19 National Diagnostic Research and Evaluation Platform. A single route to evaluate 18. new diagnostic tests for COVID-19. 2020. https://www.condor-platform.org/.

19. COVID-19 Open Access Project. Living Evidence on COVID-19. 2020. Available from: https://ispmbern.github.io/covid-19/living-review/. Accessed 14 Oct 2020.

20. Whiting PF, Rutjes AW, Westwood ME, Mallett S, Deeks JJ, Reitsma JB, et al. QUADAS-2: a revised tool for the quality assessment of diagnostic accuracy studies. Ann Intern Med. 2011;155(8):529-36.

21. Gordon AL, Franklin M, Bradshaw L, Logan P, Elliott R, Gladman JR. Health status of UK care home residents: a cohort study. Age Ageing. 2014;43(1):97-103.

Cox L, Ballantuono I, Lord J, Sapey E, Mannick J, Partridge L, et al. Tackling 22. Immunosenescence to improve COVID-19 outcomes and vaccine response in older adults. Lancet Healthy Longevity. 2020; In press.

23. Graham NSN, Junghans C, Downes R, Sendall C, Lai H, McKirdy A, et al. SARS-CoV-2 infection, clinical features and outcome of COVID-19 in United Kingdom nursing homes. J Infect. 2020;81(3):411-9.

Wang Y, Zhang D, Du G, Du R, Zhao J, Jin Y, et al. Remdesivir in adults with severe COVID-19: 24. a randomised, double-blind, placebo-controlled, multicentre trial. Lancet. 2020;395(10236):1569-78. Stowe J, Tessier E, Zhao H, Guy R, Muller-Pebody B, Zambon M, et al. Interactions between 25.

SARS-CoV-2 and Influenza and the impact of coinfection on disease severity: A test negative design. medRxiv. 2020.

26. Axell-House DB, Lavingia R, Rafferty M, Clark E, Amirian ES, Chiao EY. The Estimation of Diagnostic Accuracy of Tests for COVID-19: A Scoping Review. J Infect. 2020.

27. Wikramaratna P, Paton R, Ghafari M, Lourenco J. Estimating false-negative detection rate of SARS-CoV-2 by RT-PCR. medRxiv. 2020.

28. Kucirka LM, Lauer SA, Laeyendecker O, Boon D, Lessler J. Variation in False-Negative Rate of Reverse Transcriptase Polymerase Chain Reaction-Based SARS-CoV-2 Tests by Time Since Exposure. Ann Intern Med. 2020.

29. Woloshin S, Patel N, Kesselheim AS. False Negative Tests for SARS-CoV-2 Infection -Challenges and Implications. N Engl J Med. 2020;383(6):e38.

30. Graziadio S, Hicks T, Allen A, Suklan J, Urwin S, Winter A, et al. A composite reference standard for COVID-19 diagnostic accuracy studies: a roadmap. 2020. https://www.cebm.net/covid-19/a-composite-reference-standard-for-covid-19-diagnostic-accuracy-studies-a-roadmap/. Accessed 14 Oct 2020.

Rutjes A, Reitsma J, Coomarasamy A, Khan K, Bossuyt P. Evaluation of diagnostic tests 31. when there is no gold standard. A review of methods. 2007.

32. Umemneku Chikere CM, Wilson K, Graziadio S, Vale L, Allen AJ. Diagnostic test evaluation methodology: A systematic review of methods employed to evaluate diagnostic tests in the absence of gold standard - An update. PLoS One. 2019;14(10):e0223832.

33. Boone DC, Azen SP, Lin CM, Spence C, Baron C, Lee L. Reliability of goniometric measurements. Phys Ther. 1978;58(11):1355-60.

Declarations

Ethics approval and consent to participate

This work was based on a pooled analysis of anonymised data from two previous studies; the CoV-19POC study, described by Clark et al. (9), the "Southampton study" [ISRCTN:14966673, date registered: 18/03/2020]; and a study described by Karim et al. (13), the "Kettering study". The Southampton study was approved by the South Central - Hampshire A Research Ethics Committee: REC reference 20/SC/0138, on the 16th March 2020. The protocol is available at: https://eprints.soton.ac.uk/439309/1/CoV _ 19POC _ Protocol _ v1.1 _ eprints.pdf. The Kettering study was approved by the Kettering General Hospital Ethics Committee. Informed consent was obtained from all participants and failure to consent was considered an exclusion criterion in both studies.

Consent for publication

Informed consent was obtained from all participants.

Availability of data and materials

An anonymised minimum dataset containing enough information to reproduce the diagnostic accuracy statistics is available from the corresponding author on reasonable request. The complete anonymised dataset is not currently available as it is still being used by the Southampton study team to produce further publications.

Competing interests

MHW co-led an (unpublished) pilot study of FebriDx in 2019 for which free kits were provided by the manufacturer (Lumos). Charitable funding has been obtained to carry out an (as yet unstarted) follow on study: Clinical utility of FebriDx in determining whether or not patients presenting to a UK Accident and Emergency Department with symptoms of acute respiratory infection require antibiotic treatment (Jon Moulton Foundation (2020) - £151K (Co-Led by MHW)). The other authors declare no relevant conflicts of interest.

Funding

This study is part of the CONDOR platform (18) which is funded by the UKRI, Asthma UK and the British Lung Foundation. SGU, BCL, KG, JS, SG, DAP and AJA are supported by the National Institute for Health Research (NIHR) Newcastle In Vitro Diagnostics Co-operative. MHW is supported by the NIHR Leeds In Vitro Diagnostics Co-operative. DSL receives funding from the NIHR Community Healthcare MedTech and In Vitro Diagnostics Co-operative at Oxford Health NHS Foundation Trust. The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care.

Authors' contributions

RB and AJA devised the project. SGU, BCL and AJA designed the analysis plan. JS designed and executed the literature search. SGU and KG conducted the RoB assessment. SGU conducted the analysis with supervision from BCL and AJA. SGU drafted the initial manuscript. All authors contributed to drafts and revisions of the manuscript.

Acknowledgements

The authors would like to acknowledge Lumos Diagnostics for sharing information vital to the design of the study, however they had no part in the study design, analysis or of the development of the manuscript.

List of abbreviations

AMU	Acute Medical Unit
ARI	Acute respiratory illness
b	Regression coefficient (estimate)
CI	Chief Investigator
CI	Confidence intervals
COVID-19	Coronavirus disease 2019
CRP	C-reactive protein
ED	Emergency Department
GICU	General Intensive Care Unit
HDU	High Dependency Unit
IPD	Individual patient data
IQR	Interquartile range
LFD	Lateral flow device
MxA	Myxovirus resistance protein A
NLR	Negative likelihood ratio
NPV	Negative predictive value
р	P value
PHE	Public Health England
PLR	Positive likelihood ration
POC	Point-of-care
PPV	Positive predictive value
RT-PCR	Reverse transcriptase polymerase chain reaction
RdRp	RNA-dependent RNA polymerase
RoB	Risk of bias
SOB	Shortness of breath
UK	United Kingdom

Supplementary material

Supplementary material 1: 'Supplementary material 1.pdf' presents the STARD checklist.

Supplementary material 2: 'Supplementary material 2.pdf' presents the PRISMA-IPD checklist.

Supplementary material 3: 'Supplementary material 3.pdf' presents the minimum data set that was requested from the Chief investigators (CIs) of the included studies.

Supplementary material 4: 'Supplementary material 4.pdf' presents the methods for calculating the diagnostic accuracy measures.

Supplementary material 5: 'Supplementary material 5.pdf' presents the PRISMA-IPD flow diagram.

Supplementary material 6: 'Supplementary material 6.pdf' presents risk of bias (RoB) assessments for the included studies.

Supplementary material 7: 'Supplementary material 7.pdf' presents the flow of patients in the Southampton study.

Supplementary material 8: 'Supplementary material 8.pdf' presents the flow of patients in the Kettering study.